Author’s response to reviews

Title: "I struggle to count my blessings ". Recovery after hip fracture from the patients’ perspective.

Authors:

Vigdis Bruun-Olsen (vigdis.bruun-olsen@vestreviken.no)
Astrid Bergland (astrid.bergland@hioa.no)
Kristi Elisabeth Heiberg (Kristi.Elisabeth.Heiberg@vestreviken.no)

Version: 3 Date: 11 Jan 2018

Author’s response to reviews:

Response letter

Date: 11.01.2018

Dear Editor,

Thank you for your valuable comments helping us to further improve our manuscript, BGTC-D-17-00419: “I struggle to count my blessings”. Recovery after hip fracture from the patients’ perspective

The answers to your requests are listed below

1. The ages previously included after the quotes are removed throughout the manuscript.

2. We have also removed the marital status in table 1 and replaced “type of fracture” with “type of procedure”.

3. We have re-phrased some of the sentences in the background, methods and research design sections to minimise textual overlaps.

4. Reference nr.28 is removed from the main manuscript. The reference list is adjusted accordingly.

We look forward to hearing from you.

Yours sincerely,
Vigdis Bruun-Olsen (RPT, PhD)

Corresponding Author
vigdis.bruun-olsen @vestreviken.no